Author’s response to reviews

Title: Mental health: A cause or consequence of injury? A population-based matched cohort study

Authors:

Cate M Cameron (cate.cameron@griffith.edu.au)
David M Purdie (davidpurdie@fastmail.fm)
Erich V Kliewer (erich.kliewer@cancercare.mb.ca)
Rod J McClure (r.mcclure@griffith.edu.au)

Version: 3 Date: 15 April 2006

Author’s response to reviews:

Thank you for your email agreeing to accept my manuscript for publication subject to formatting changes. All changes have been completed as requested.

Formatting changes requested

------------------------------
CHANGED Word count Please remove this from the manuscript.

CHANGED Keywords section Please remove the keywords section from the manuscript.

CHANGED Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT.

CHANGED References - In the manuscript text, please place the reference before the full stop.

CHANGED Figure and table placement text Please remove the figure and table placement text. Figures and tables will be included in the final version of the manuscript as near as possible to their first mention in the manuscript.

CHANGED Competing interests - If there are none to declare, please write only The authors declare that they have no competing interests.

CHANGED Please move the tables to after the references.

------------------------------